Author’s response to reviews

Title: Nutrition in primary and secondary prevention of cardiovascular risk in the continental and Mediterranean regions of Croatia

Authors:

Jozica Sikic (josicas1@gmail.com)

Mira Stipcevic (mira.stipcevic@gmail.com)

Hrvoje Vrazic (vrazic@gmail.com)

Jasna Cerkez Habek (jasna.habek@gmail.com)

Eduard Margetic (eduard.margetic@gmail.com)

Dario Gulin (dariogulin@gmail.com)

Version: 1 Date: 26 Apr 2017

Author’s response to reviews:

Dear Editor,

Improvements to the English language were provided by English language editing service as it has been requested. In revised manuscript, it was more explained in the methodology section about defining acute vs. chronic coronary heart disease. Also, new information about diet index including reproducibility of the FFQ were added. The tables were embedded in the word document.